Author's response to reviews

Title: Plasma Chemokine Signature Correlates with Lung Goblet Cell Hyperplasia in Smokers with and without Chronic Obstructive Pulmonary Disease

Authors:

Victor Kim (victor.kim@tuhs.temple.edu)
William D Cornwell (cornwell@temple.edu)
Michelle Oros (Michelle.grant@tuhs.temple.edu)
Heba Durra (heba.durra@tuhs.temple.edu)
Gerard J Criner (gerard.criner@tuhs.temple.edu)
Thomas J Rogers (thomas.rogers@temple.edu)

Version: 4 Date: 14 September 2015

Author’s response to reviews:

1. Clarify if consent as written or oral.

Response: This was clarified in the manuscript.

2. Clarify in the Authors' Contribution Section that all authors read and approved the final version.

Response: This was clarified in the authors’ contribution section.